Author's response to reviews

Title: Motives for (not) participating in a lifestyle intervention trial

Authors:

Jeroen Lakerveld (j.lakerveld@vumc.nl)
Wilhelmina IJzelenber (helma.ijzelenberg@vumc.nl)
Maurits W van Tulder (mw.vantulder@vumc.nl)
Irene M Hellemans (i.hellemans@vumc.nl)
Jan A Rauwerda (ja.rauwerda@vumc.nl)
Albert C van Rossum (ac.vrossum@vumc.nl)
Jaap C Seidell (jaap.seidell@falw.vu.nl)

Version: 4 Date: 7 April 2008

Author's response to reviews:

Dear BioMed Central Editorial Production Team,

We have made the formatting changes as requested:
- The reference list contain all authors' names, and the term 'et al.' is not being used anymore.
- The Tables are now referred to in standard numerals instead of Roman Numerals.

These changes are made in the latest version of the manuscript, as provided by you in the e-mail.

Please do not hesitate to contact me if there remain formatting changes to be made.

Sincerely,

Jeroen Lakerveld,
corresponding author